Author’s response to reviews

Title: Lack of Association of the Opioid Receptor mu 1 (OPRM1) A118G Polymorphism (rs1799971) with Alcohol Dependence: Review and Meta-analysis of Retrospective Controlled Studies

Authors:

Xiangyi Kong (Xiangyi_Kong@hms.harvard.edu)

Hao Deng (XKONG@mgh.harvard.edu)

Shun Gong (gongshunsmmu@foxmail.com)

Theodore Alston (Alstoncorbett@comcast.net)

Yanguo Kong (kongyanguo1@gmail.com)

Jingping Wang (JWANG23@mgh.harvard.edu)

Version: 2 Date: 26 Sep 2017

Author’s response to reviews:

Dear editor,

Thank you for your comments. We have revised the article point by point. The main changes and the responses to the reviewers are as follows:

MGTC-D-17-00085R1

Association between Opioid Receptor mu 1 (OPRM1) A118G Polymorphism (rs1799971) and Alcohol Dependence

BMC Medical Genetics

Reviewer reports:

Renato Polimanti, Ph.D. (Reviewer 2): The authors addressed my comments. However, the new literature references included in the manuscript were not included in the reference list. The authors should check their references carefully. I also suggest to include the reference number in the literature cited in the tables to facilitate the readers.
Corrected as suggested.

David Jeffries (Reviewer 3): The authors have adequately addressed my comments

Many thanks!

Best wishes,

Yanguo Kong